Author’s response to reviews

Title: Effectiveness of a training program on compliance with recommendations for venous lines care

Authors:

MARIA JESUS PEREZ-GRANDA (massus@hotmail.es)
MARIA GUEMBE (mariaguembe@hotmail.com)
CRISTINA RINCON (crisrsanz@yahoo.es)
PATRICIA MUÑOZ (pmunoz@hggm.es)
EMILIO BOUZA (emilio.bouza@gmail.com)

Version: 6 Date: 9 April 2015

Author’s response to reviews: see over
Dr Walter Zingg

e-mail: editorial@biomedcentral.com
Subject: MS: 1931586571157398
Date: April 9, 2015

Dear Dr. Zingg,

Thank you very much for your comments and help about our article entitled “Impact of a training program on the care of peripheral and central lines: assessment with 2 point prevalence studies”. We proceed to answer all the questions.

Line 89: Remove "comprised"

This is now been removed in the text.

Line 105: Remove "o"

This is now been removed in the text.

Lines 119-122: English

We have reviewed the English of this text as suggested.

Line 123: What does "Adequate correct care items" mean?

This is now been removed "Adequate correct care items" because really we mean when a line was considered adequate.

Line 143: Please provide the name of the "median" test (e.g. ranksum test) Line 183: You may remove my name (it is not necessary and the only part of the manuscript you do so)

We have included “Kruskal-Wallis test" in the Statistical analysis section.

Thank you once more for your kind attention to our manuscript and all your help.

Maria Jesus Pérez
Clinical Microbiology and Infectious Diseases
Hospital General Universitario Gregorio Marañón
Doctor Esquerdo, 46. 28007 Madrid, Spain

Phone: 34-91-5868027; Fax: 34-91-5044906

E-mail: massus@hotmail.es